Author’s response to reviews

Title: Opiophobia: the situation among the general population and health care professionals in North-Eastern Portugal

Authors:

Henk Verloo (henk.verloo@hcuge.ch)
Emmanuel K Mpinga (emmanuel.kabengele@unige.ch)
Maria Ferreira (mariaferreira77@hotmail.com)
Charles-Henri Rapin (ch-rapin@unige.ch)
Philippe Chatonay (philipee.chastonay@unige.ch)

Version: 2 Date: 29 May 2009

Author’s response to reviews:

1. Adaption of the abstract
2. We adapted the manuscript with the statement of the approval of an appropriate ethics committee
3. We translated the two questionnaires in English